Author's response to reviews

Title: Pulmonary fibrosis secondary to siderosis causing symptomatic respiratory disease: a case report

Authors:

Liam M McCormick (drliammccormick@hotmail.com)
Martin Goddard (Martin.Goddard@papworth.nhs.uk)
Ravi Mahadeva (rm232@cam.ac.uk)

Version: 3 Date: 16 April 2008

Author's response to reviews: see over
To the Editors of the Journal of Medical Case Reports,

Many thanks for considering our case report for submission. In response to the specific comments made by the reviewers:

**Reviewer 1**

We have now included a photo demonstrating significant pulmonary fibrosis (see Fig 3(B) - High power Elastic van Gieson collagen stain (red) demonstrating significant fibrosis).

**Reviewer 2**

An experienced Histopathologist (MG) has reviewed the histopathology. He did not identify any granulomas in the lung tissue.

We hope that the paper is now suitable for publication.

Many thanks,

Liam McCormick, Ravi Mahadeva

Department of Respiratory Medicine,

Addenbrooke’s Hospital,

Cambridge, UK